Author's response to reviews

Title: A randomised controlled trial of preventive spinal manipulation with and without a home exercise program for patients with chronic neck pain.

Authors:

Johanne Martel (johanne.martel@uqtr.ca)
Claude Dugas (claude.dugas@uqtr.ca)
Jean-Daniel Dubois (jean-daniel.dubois@uqtr.ca)
Martin Descarreaux (martin.descarreaux@uqtr.ca)

Version: 6 Date: 19 January 2011

Author's response to reviews:

Dear M.Aulakh,

Please find included our latest revised version of the manuscript entitled "A randomised controlled trial of preventive spinal manipulation with and without a home exercise program for patients with chronic neck pain". The comment made by the reviewer (Dr Peloso) has been addressed and report of the pain outcome as been standardised in all tables and figures. These changes are highlighted in yellow.

Response to reviewer: As suggested by the reviewer the term VAS as been modified to Mean pain (VAS) in table 2-3 as well as in figure 2.

Sincerely,

Martin Descarreaux